Author's response to reviews

Title: Attitudes towards Terminal Sedation: An Empirical Survey among Experts in the Field of Medical Ethics

Authors:

Alfred Simon (asimon1@gwdg.de)  
Magdalene Kar (m.kar@arcor.de)  
Jose Hinz (jhinz@zari.de)  
Dietmar Beck (dbeck@zari.de)

Version: 3 Date: 1 April 2007

Author's response to reviews: see over
Cover letter for final version

MS: 1213897101117996

Attitudes towards terminal sedation: an empirical survey among experts in the field of medical ethics. Alfred Simon, Magdalene Kar, Jose Hinz and Dietmar Beck

Formatting changes requested

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).
Done.

Please include the city and country in the affiliations.
Done.

References - References should be cited in the text using consecutive numbers in square brackets starting from [1] etc. Please ensure the type of bracket used is consistent. Please do not use superscript numbers.
Done.

References - In the manuscript text, please place the reference before the full stop.
Done.

Website addresses - Websites must be removed from the text, and cited from the main reference list only. If you are using referencing software such as EndNote or Refman, please ensure that your references are entered through the software rather than included in the manuscript reference list manually. Please present your reference in the following structure, e.g. 17. Website title [www.website.com].
Done.

References - The reference list should contain all authors’ names, if there are 30 or below. The term 'et al.' should not be used.
Done.

Acknowledgements - Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.
No funding.

With best regards

Alfred Simon
(On behalf of the authors)